Author’s response to reviews

Title: The Alu-insertion progesterone receptor gene polymorphism is not associated with breast cancer: a meta-analysis

Authors:

Jun Yao (yaojun198717@163.com)

Xing-ling Qi (995140327@qq.com)

Yong Zhang (zhangyong@cancerhosp-ln-cmu.com)

Version: 3 Date: 15 Jan 2018

Author’s response to reviews:

Response to Editor Comments:

Comment 1. Please include the email addresses of all authors on the title page.

Answer. We have added the email addresses of all authors on the title page. Please check it.

Comment 2. Please change the Abstract headings to: Background, Methods, Results, Conclusions.

Answer. We have revised the Abstract in the revised manuscript. Please check it.

Comment 3. Please change the main text headings to: Background, Methods, Results, Discussion, Conclusions.

Answer. We have revised the manuscript according to your suggestion. Please check it.

Comment 4. Please represent authors’ names using their initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Answer. We have represented the authors’ names using their initials in the Authors’ Contributions section. Please check it.
Comment 5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer. We have uploaded the required submission according to your suggestion.